Author's response to reviews

Title: Knockout of MIMP protein in Lactobacillus plantarum lost its regulation of intestinal permeability on NCM460 epithelial cells via the zonulin pathway

Authors:

Zhihua Liu (liuzhh26@mail.sysu.edu.cn)
Liang Kang (xwz1982@live.cn)
Chao Li (chaozs6y@outlook.com)
Chao Tong (1527111808@qq.com)
Meijin Huang (mjhuang@live.cn)
Xingwei Zhang (lkang1972@hotmail.com)
Nanqi Huang (nanqihuang@hotmail.com)
Mary Pat Moyer (mpmoyer@incell.com)
Huanlong Qin (hqlin@live.cn)
Jianpin Wang (j-p-wang@hotmail.com)

Version: 3
Date: 14 September 2014

Author's response to reviews: see over
Dear Dr. Magdalena Morawska,

Thank you for your advice.

We would like to re-submit our manuscript, which has been revised according to the comments. We have addressed each comment raised, and revised the paper accordingly. All amendments are highlighted in red font in the revised manuscript. Point-by-point replies are included below. Furthermore, this manuscript has been edited and proofread. We hope that the revised manuscript is now acceptable for publication in your journal.

We look forward to hearing from you soon.

Sincerely yours,

Zhihua Liu, Ph.D., M.D.
Department of Colorectal Surgery, the Sixth Affiliated Hospital of Sun Yat-sen University.

liuzhh26@mail.sysu.edu.cn
Response to Reviewers’ Comments:

Editorial Requirements:

**request for copy editing***

We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Answer: Thank you for your comments. Our manuscript has been edited and proofread by professional editors (a native English speaking colleague). We hope that the revised language is now acceptable for publication.